Author's response to reviews

Title: Overuse of non-prescription analgesics by dental clinic patients

Authors:

Kennon J Heard (kennon.heard@rmpdc.org)  
Nicole L Ries (nlries@gmail.com)  
Richard C Dart (Richard.dart@rmpdc.org)  
Gregory M Bogden (gregory.bogdan@rmpdc.org)  
Richard D Zallen (Richard.zallen@dhh.org)  
Frank Daly MBBS (frankdaly@mac.com)

Version: 7 Date: 8 December 2008

Author's response to reviews:

Abstract: Introduction removed the phrase "and perceptions of non-prescription analgesic safety.

Introduction: Changed ""While patients with occult acetaminophen overdose may present with acute liver failure, there are also risks... " to

"Patients with occult acetaminophen overdose present with acute liver failure and there are also risks "